Dear editor and reviewers:

Thank you for your letter and the reviewers’ comments on our manuscript entitled “3D time-of-flight magnetic resonance angiography combined with high resolution T2-weighted imaging in preoperative evaluation of microvascular decompression: A meta-analysis” (ID: 79933). Those comments are very helpful for revising and improving our paper, as well as the important guiding significance to other studies. We have studied the comments carefully and made corrections which we hope to meet with approval. The main corrections are in the manuscript and the responses to the reviewers’ comments are as follows (the replies are highlighted in blue).

Replies to the reviewers’ comments:

Reviewer #1:

This is a good study. The version of PRISMA in use is old. Download and use PRISMA 2020, and update the whole manuscript in accordance with it. Upload the filled out PRISMA 2020. Also change the PRISMA figure of the paper into PRISMA 2020 figure. The paper is good but I am not sure if a meta-analysis is good for this particular journal’s scope.

1. The abstract and text have been revised according to the PRISMA 2020 checklist.
2. The PRISMA figure used in this meta-analysis is the PRISMA figure 2020 version.

Reviewer #2:

I would like to thank the authors for their excellent manuscript. It is well written and a well performed study. The topic is relevant for daily patient care. I have no comments to add.

Thanks for your comments.

Replies to the Company editor-in-chief:
Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. In order to respect and protect the author’s intellectual property rights and prevent others from misappropriating figures without the author's authorization or abusing figures without indicating the source, we will indicate the author's copyright for figures originally generated by the author, and if the author has used a figure published elsewhere or that is copyrighted, the author needs to be authorized by the previous publisher or the copyright holder and/or indicate the reference source and copyrights. Please check and confirm whether the figures are original (i.e. generated de novo by the author(s) for this paper). If the picture is ‘original’, the author needs to add the following copyright information to the bottom right-hand side of the picture in PowerPoint (PPT):

The figures have been processed as required.

Authors are required to provide standard three-line tables, that is, only the top line, bottom line, and column line are displayed, while other table lines are hidden. The contents of each cell in the table should conform to the editing specifications, and the lines of each row or column of the table should be aligned

The tables have been processed as required

Before final acceptance, when revising the manuscript, the author must supplement and improve the highlights of the latest cutting-edge research results, thereby further improving the content of the manuscript. To this end, authors are advised to apply a new tool, the RCA. RCA is an artificial intelligence technology-based open multidisciplinary citation analysis database. In it, upon obtaining search results from the keywords entered by the author, "Impact Index Per Article" under "Ranked by" should be selected to find the latest highlight articles, which can then be used to further improve an article under preparation/peer-review/revision
We use the title or keyword to search in RCA, but unfortunately no relevant literature has been retrieved.